Author's response to reviews

Title: Modified versus standard intention-to-treat reporting: Are there differences in methodological quality, sponsorship, and findings in randomized trials? A cross-sectional study.

Authors:

Alessandro Montedori (amontedori@regione.umbria.it)
Maria Isabella Bonacini (isabella.bonacini@tiscali.it)
Giovanni Casazza (giovanni.casazza@unimi.it)
Maria Laura Luchetta (lauramary@virgilio.it)
Piergiorgio Duca (piergiorgio.duca@unimi.it)
Francesco Cozzolino (ankom@interfree.it)
Iosief Abraha (iosief_a@yahoo.it)

Version: 3 Date: 20 February 2011

Author's response to reviews:

Editor,

thank you for accepting the manuscript.

The following are the changes we made including those suggested by you:

the corresponding author is now the first author,
the Box was converted to Appendix,
reference 21 is corrected,
and few other corrections.

Kind regards.

Iosief Abraha